Author's response to reviews

Title: Do intrauterine or genetic influences explain the foetal origins of chronic disease? A novel experimental method for disentangling effects.

Authors:

Anita Thapar (thapar@cf.ac.uk)
Gordon Harold (harold@cf.ac.uk)
Frances Rice (ricef@cf.ac.uk)
XiaoJia Ge (xjge@ucdavis.edu)
Jacky Boivin (boivin@cf.ac.uk)
Dale Hay (haydf@cf.ac.uk)
Marianne van den Bree (vandenbreemb@cf.ac.uk)
Allyson Lewis (lewisa@cf.ac.uk)

Version: 5 Date: 21 June 2007

Author's response to reviews:

figure legend inserted
tables-inerted
conflict of interest statement changed